**Author’s response to reviews**

**Title:** Sexual and reproductive health of women living with HIV in Muslim-majority countries: A systematic mixed studies review

**Authors:**

Dyah Juliastuti (d.juliastuti@uq.edu.au)

Judith Dean (j.dean4@uq.edu.au)

Lisa Fitzgerald (l.fitzgerald@sph.uq.edu.au)

**Version: 6 Date: 26 Feb 2020**

**Author’s response to reviews:**

Dear Editor,

I have revised the manuscript per your suggestions, including changing abstract structure, details of author's contributions, and some typing errors. Thank you for the feedback.

Kind regards,

Dyah Juliastuti

Dr Lisa Fitzgerald

Dr Judith Dean